\chapter{Visual Persistence of Used Online Sources}
The following images provide insight into the online sources referenced in this work:

\begin{figure}[H]
  \centering
  \includegraphics[width=15cm]{online/cobilab.png}
  \caption{View of Cobilabs - creators of \acs{GeCo} - github page \cite{geco-repo}}
  \label{}
\end{figure}

\begin{figure}[H]
  \centering
  \includegraphics[width=15cm]{online/ensembl.png}
  \caption{View of ensembl genome browser splash page \cite{ensembl}.}
  \label{}
\end{figure}


\begin{figure}[H]
  \centering
  \includegraphics[width=15cm]{online/ftp-1000genomes.png}
  \caption{View of the browser view for IGSRs \acs{FTP}-Server \cite{ftp-igsr}.}
  \label{}
\end{figure}


\begin{figure}[H]
  \centering
  \includegraphics[width=15cm]{online/ftp-ensembl.png}
  \caption{View of the browser view for Ensembls \acs{FTP}-Server \cite{ftp-ensembl}.}
  \label{}
\end{figure}


\begin{figure}[H]
  \centering
  \includegraphics[width=15cm]{online/ftp-ncbi.png}
  \caption{View of the browser view for NCBIs \acs{FTP}-Server \cite{ftp-ncbi}.}
  \label{}
\end{figure}


\begin{figure}[H]
  \centering
  \includegraphics[width=15cm]{online/gpl.png}
  \caption{View of the GNU Public License page \cite{gpl}.}
  \label{}
\end{figure}


\begin{figure}[H]
  \centering
  \includegraphics[width=15cm]{online/mit-license.png}
  \caption{View of MIT license description \cite{mitlic}.}
  \label{}
\end{figure}

\begin{figure}[H]
  \centering
  \includegraphics[width=15cm]{online/samtools_spec.png}
  \caption{View of Samtools sourcefiles and file format description \cite{bam}.}
  \label{}
\end{figure}

\begin{figure}[H]
  \centering
  \includegraphics[width=15cm]{online/ucsc.png}
  \caption{View of UCSC splash page \cite{ucsc}.}
  \label{}
\end{figure}

